Dear editor and reviewers,

We really appreciate you for your carefulness and conscientiousness. Your suggestions are really valuable and helpful for revising and improving our manuscript titled “Castleman Disease of the Pancreas Mimicking Pancreatic Malignancy on 68Ga-DOTATATE and 18F-FDG PET/CT: A case report”. According to your suggestions, we will make the following revisions on this manuscript and upload the file in the format mentioned by the editor as soon as possible:

Reviewer #1: Has the document been approved by the hospital ethics Committee? Please add this content. Was the manuscript given informed consent by the patient? Please add this content. Please provide a copy of the first page of the patient’s inpatient case.

Response to reviewer #1: Thank you very much for your support and advice. We will add the files mentioned as soon as possible.

Reviewer #2: Specific comments on weaknesses of the article and what could be improved:

Major points - none

Minor points - what are the clinical implications and recommendations based on this case series? What is the prognosis?

Response to reviewer #2: Thank you for your positive comments. For the shortcomings of this article, we will add relevant clinical significance, recommendations and prognosis based on this case series.

We hope that the revised version of the manuscript will be acceptable for publication in World Journal of Gastrointestinal Surgery. If you have any queries, please do not hesitate to contact us.

Thank you again for your valuable comments and suggestions. I look forward to hearing from you soon in due course.

Yours sincerely,
Kai He